

May 10, 2015

Dear Editor,

Please find enclosed the edited manuscript in Word format (file name: 17353-review.doc).

Title: Intraperitoneal Tuberculous Abscess: CT Features

Author: Peng DONG, Jing-jing Chen, Xi-zhen WANG, Ya-qin WANG

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 17353

The manuscript has been improved according to the suggestions of reviewers:

1 Format has been updated

2 Revision has been made according to the suggestions of the reviewer

(1) Page 6: the word "continuitatem" should be "contiguity".

Revision has been made.

(2) Page 11: the word "iamges" is misspelled.

Revision has been made.

(3) Please provide language certificate letter by professional English language editing companies.

Language certificate letter has been provided.

(4) Please rearrange all the authors' affiliations with Department, University or Institute, City, Postcode, Country, etc. Please see the format in the attachment (without any symbol or figure like * or 1).

All the authors' affiliations has been rearranged.

(5) Please provide the author contributions. Authors must indicate their specific contributions to the published work.

Author contributions has been provided.

(6) Please reformat all the reference numbers (superscript with square brackets). Please check throughout. Normal line space is required. Please check throughout.

All the reference numbers has been reformatted. The authors have checked all the reference numbers and the normal line space.

(7) Ethics approval, Biostatistics, Informed consent, Conflict-of-interest, Data sharing. Authors are required to make these statements in the manuscript's title page. A copy of any approval document(s)/letter(s) or waiver of confirmation must also be provided in PDF format. BPG will include all the confirmations along with the manuscript as a permanent part of the online publication.

The statements (Ethics approval, Biostatistics, Informed consent, Conflict-of-interest, Data sharing) have been provided. The authors write the statements in the manuscript's title page.

(8) Correspondence to: Corresponding Author is the person responsible for making any edits/submitting revisions to the manuscript and is the only author connected to the manuscript who may view the progress of the manuscript as it moves from one stage to the next. Please provide the institution more detailed here, just like corresponding author, professional title,

Department, University, detailed address, City, Post Code, Country. Please provide these files needed for retrospective study, each in a separate PDF file, signed by the Correspondence author or a copy of Institution approval document(s)/letter(s) or waiver of confirmation. For sample wording and detailed information, please see the Revision policy in the attachment or Instruction to authors on our website.

More detailed information about the corresponding author has been provided.

- (9) Structured abstract: An informative, structured abstract of no less than 336 words should accompany each manuscript. The Abstract will be structured into the following sections and adhering to the word count thresholds indicated in parentheses: AIM (no more than 20 words): The purpose of the study should be stated clearly and with no or minimal background information, following the format of: "To investigate/study/determine..." METHODS (no less than 140 words): You should present the materials and methods used for all of the data presented in the proceeding Results section of the abstract. RESULTS (no less than 150 words): Observational and experimental results, including data, effects and outcomes, should be presented. You should present P values where appropriate. You must provide relevant data to illustrate how the statistical values were obtained, e.g. 6.92 ± 3.86 vs 3.61 ± 1.67 , $P < 0.001$. Articles not meeting this requirement will not be accepted. CONCLUSION (no more than 26 words): You should provide an analysis and evaluation of the hypothesis and your results, along with the implications of your findings that fit within the scope of your data. The conclusion should be written in the present tense.

The Abstract has been provided.

- (10) Key words: A list of 5–10 keywords should be given below the abstract, each separated by a semicolon (;). Whenever possible, the terms should be from the HYPERLINK "<http://www.nlm.nih.gov/mesh/MBrowser.html>" Medical Subject Headings list of Index Medicus. Core tip: Please write a summary of less than 100 words to outline the most innovative and important arguments and core contents in your paper to attract readers.

Key words have been provided.

- (11) Audio Core Tip: In order to attract readers to read your full-text article, we request that the first author make an audio file describing your final core tip. This audio file will be published online, along with your article. Please submit audio files according to the following specifications: Acceptable file formats: .mp3, .wav, or .aiff. Maximum file size: 10 MB. To achieve the best quality, when saving audio files as an mp3, use a setting of 256 kbps or higher for stereo or 128 kbps or higher for mono. Sampling rate should be either 44.1 kHz or 48 kHz. Bit rate should be either 16 or 24 bit. To avoid audible clipping noise, please make sure that audio levels do not exceed 0 dBFS.

Audio Core Tip has been provided.

- (12) Statistical analysis: Statistical methods should be described when they are used to verify the results.

The authors do not use the statistical analysis. The reason has been provided.

- (13) Provide the "Highlighted contents".

Highlighted contents have been provided.

- (14) Tables shouldn't be images, but as word or excel format so that I can edit them easily.

The format of the tables was corrected.

3 References and typesetting were corrected

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

DONG Peng

Medical Imaging Centre of the Affiliated Hospital,

Weifang Medical University,

Shenglidong Street, Weifang 261031,

Shandong Province, China

Telephone: +86-536-8462397

E-mail: dongpeng98021@sina.com